Author’s response to reviews

Title: Stigma in health facilities: why it matters and how we can change it

Authors:

Laura Nyblade (lnyblade@rti.org)
Melissa Stockton (stocktonmelissa@gmail.com)
Kayla Giger (kmg110@georgetown.edu)
Virginia Bond (GBond@zambart.org.zm)
Maria Ekstrand (Maria.Ekstrand@ucsf.edu)
Roger Mc Lean (rmcleanuwi@yahoo.com)
Ellen Mitchell (ellenmaevemitchell@gmail.com)
LaRon Nelson (Laron_Nelson@urmc.rochester.edu)
Jaime Sapag (Jaime.Sapag@camh.ca)
Taweesap Siraprapasiri (taweesap@rocketmail.com)
Janet Turan (jmturan@uab.edu)
Edwin Wouters (edwin.wouters@uantwerpen.be)

Version: 2 Date: 20 Nov 2018

Author’s response to reviews:

Please see attached cover letter.